Author's response to reviews

Title: Efficacy and safety of moxifloxacin in acute exacerbations of chronic bronchitis: a prospective, multicenter, observational study (AVANTI)

Authors:

Alexander Chuchalin (chuchalin@inbox.ru)
Maryna Zakharova (ZakharovaMarina@i.ua)
Dejan Dokic (drdejand@yahoo.com)
Mahir Tokić (asim_tokic@yahoo.com)
Hans-Peter Marschall (hans-peter.marschall@bayer.com)
Thomas Petri (thomas.petri@bayer.com)

Version: 4 Date: 16 January 2013

Author's response to reviews:

Dear Sir/Madam!

The requested information about the study approval by the national ethical commitees was added as requested. A list of all investigators is available, I will get in contact by phone with the editor on the use of this list.

Best regards

Thomas Petri